Author’s response to reviews

Title: Using a Computational Model to Quantify the Potential Impact of Changing the Placement of Healthy Beverages in Stores as an Intervention to "Nudge" Adolescent Behavior Choice

Authors:

Michelle Wong (mwong23@jhu.edu)
Claudia Nau (Cnau1@jhu.edu)
Anna Kharmats (Akharma1@jhu.edu)
Gabriela Vedovato (Gabriela.vedovato@uifesp.br)
Lawrence Cheskin (cheskin@jhu.edu)
Joel Gittelsohn (jgittel1@jhu.edu)
Bruce Lee (brucelee@jhu.edu)

Version: 2 Date: 11 Dec 2015

Author’s response to reviews:

Editors’ comment: Please ensure that all authors of the manuscript are referred to in the authors contributions section. Please also use the initials of each author in this section and not their full name.

Response: We have changed the Authors’ Contribution section (page 11 – 12, lines 256 – 259) as follows:

“Authors’ contributions: MSW, CN, AYK, LJC, and BYL developed the initial research question and model. AYK, GMV, and JG assisted with model population and calibration. All authors helped analyze and interpret the results, contributed to the writing and revising of the manuscript, and approved the final manuscript.”